Author's response to reviews

Title: A new method to assess perceived well-being among elderly people - a feasibility study

Authors:

Jori S Reijula (jori.reijula@tkk.fi)
Toni Rosendahl (toni.rosendahl@gmail.com)
Kari E Reijula (kari.reijula@ttl.fi)
Paula Roilas (paula.roilas@terveyspalvelu.fi)
Heikki Roilas (heikki.roilas@lappeenranta.fi)
Raimo Sepponen (raimo.sepponen@tkk.fi)

Version: 6 Date: 18 November 2009

Author's response to reviews:

Dear Editor,

Thank you very much for the reviewer's comment. I have now made the corrections requested:

Question (Referee 1): Since the RAVA is a national measuring instrument, the range of the index values of the measure should be given

Answer: I clarified the RAVA index score in the sentence page 6 line 6.

Thank you very much for accepting my manuscript for publication and thank you very much for the reviewers for giving me advice with my manuscript!

Sincerely,

Jori Reijula